Shi Jin ( )

Brief CV

Peking University University of Arizona. C. David Levermore , George C.

Papanicolaou

Visiting Positions

Feng Kang Prize for Scientific Computing, , 2001--, Distinguished Overseas Young Researcher Award (Class B)

Chief Editor of Communications in Mathematical Sciences,

Teaching

Spring 2004:

Previous Semesters Ph.D Students:

Xiantao Li Computational Mathematics, Princeton University Xuelei Wang Current position: Software Engineer at Verizon Xiaomei Liao Liming Lin Longqing Wu

Research

listed by subjects listed by years of publication listed in AMS MathSciNet and Colloquia MathSciNet Science Citation Index

The HYKE Network:

3rd Annual Intermountain/Southwest Conference on Industrial and Interdisciplinary Mathematics,

Workshop on Hyperbolic Conservation Laws, International Conference on Mechanics and PDEs, -- on the occasion of Prof. Marshall Slemrod’s 60th birthday International Conference on Nonlinear Evolutional PDEs, SIAM Conference on Mathematical Aspects of Material Science Los

Papanicolaou’s 60th Birthday,

George Papanicolaou’s 60th Birthday , Frontiers in Applied Mathematics,

International Conference on Numerical and Applied PDEs, The The Sixth International Workshop on Mathematical Aspects of Fluid and IMA Thematic Year on Mathematics of Materials and Macromolecules,

Hong Kong,

Last updated 2002/01/08